





PROJECT END REPORT Clim-ATIC Work Package 4: Emergency Population Warning System

"Developing and testing a people-centred system of early warnings for extreme weather events, by the use of modern technology"



Contacts:

Mr. Haavard Stensvand, Project Manager Head of Civil Protection and Emergency Planning County Governor of Sogn og Fjordane (0047) 57 65 51 11 / (0047) 915 77 432 hst@fmsf.no

Mr. Svein Helge Steinsåker Advisor, Civil Protection and Emergency Planning County Governor of Sogn og Fjordane <u>shs@fmsf.no</u> (0047) 57 65 51 16 / (0047) 908 95 078

Abstract:

The main objective of the Clim-ATIC WP4 "Emergency Population Warning System Project" has been to demonstrate an effective, reliable and cost-efficient early warning system with a multi-hazard approach. This report argues that the technical aspects of such systems are at large readily available, whereas issues concerning confidentiality legislation and system regulations are challenges that must be addressed before successfully implementing a location-based early warning system.

	Table of contents:	
Introduction		p. 3
Chapter 1	Project context	
•	1.1 Project background	p. 4
	1.2 Adaptation outcome	p. 5
Chapter 2	The warning system	
•	2.1 Project reference group	p. 6
	2.2 System specification	p. 8
Chapter 3	Warning system suppliers	
	3.1 Mapping the landscape	p. 9
	3.2 Unified Messaging Systems	p. 9
	3.3 Fram Web	p. 10
	3.4 Choosing the system supplier	p. 10
Chapter 4	Population warning exercise, June 2 nd 2010	
•	4.1 Exercise context	p. 11
	4.2 Planning	p. 12
	4.3 Dissemination	р. 13
	4.4 The demonstration exercise	p. 14
Chapter 5	Exercise evaluation	
•	5.1 Fram Web's logging system	p. 18
	5.2 The online evaluation form	p. 20
	5.3 The door-to-door survey	p. 22
Chapter 6	Project summary and concluding remarks	
•	6.1 The warning system	p. 24
	6.2 The system operator	р. 24
	6.3 Project relevance and follow-ups	p. 25
A		
Appendix		р. 26

Total number of pages: 34

Introduction

Societies have a long record of adapting to the impacts of weather and climate change, and wide arrays of adaptation measures are currently available (*see e.g. appendix 1*).

However, due to the reach and scope of expected climate change, more extensive adaptation strategies are needed, and these strategies must be embedded within broader initiatives and trends in society. Hence, climate adaptation initiatives such as risk reduction strategies must mirror the dynamics of modern technologies in order to convey intelligent and reliable information.

The very success of a people-centred system of early warning is thus linked to the use of modern technologies to convey timely and understandable instructions to those at risk

The overall aim of the project is to demonstrate a phone-based system of early warnings *, by disseminating warning messages to all fixed phone lines and mobile phones in a specified geographic area. This phone-based early warning system will also be supplemented by other sources of information, such as Twitter and Facebook.

This report finalizes the *Emergency Population Warning System* project, and consists of six main chapters:

- Chapter one provides the project context and specifies the project background and desirable adaptation outcomes.
- In the second chapter, the project reference group members are presented, and the chapter illustrates the process of defining the technical specification and framework for the multi-hazard warning system.
- Chapter three outlines the process of identifying warning system providers which complied with the technical specification that was defined by the project reference group.
- The fourth chapter gives an account of the project's demonstration exercise: a population warning exercise in the municipality of Aurland in Sogn og Fjordane.
- Chapter five presents the evaluation of the demonstration exercise, whereas chapter six summarizes the report.

* In some contexts, population warning systems are defined as robust systems (e.g. sirens horns) that operate irrespective of commercial infrastructure, whereas this report extends the definition to incorporate commercial phone networks.

Chapter 1 Project context

1.1 Project background

Clim-ATIC is an EU-project (2008-2011) which deals with communities' ability to develop adaptation capacities to face a future climate, where the frequency and scale of extreme weather is believed to increase.

Work package 4 (WP4) in Clim-ATIC relates to the implementation of adaptation demonstration projects, and one of these WP4-projects is located in the County of Sogn og Fjordane.

The main purpose of this particular project has been to develop and demonstrate a reliable and costeffective people-centred system of early warnings for extreme weather events. A detailed project plan was formally approved by the steering group in October 2008 (*appendix 2*).

Critically, a modern population warning system must reach everyone in the specific geographic area of concern, i.e. both residents and temporary visitors.

Two of the main challenges of the project were thus interlinked:

- Defining a technical specification and a system framework which enabled cost-efficient and reliable issuing of multi-hazard warnings
- Identifying system providers which complied with the technical specifications of the above mentioned criteria

The *Emergency Population Warning System* project has been mentioned in a proposition report by the National Defence Committee (*appendix 3*), which discussed the need for a modern system for population warning. Furthermore, the majority of the committee stated that efficient population warning by the use of modern technology constitutes an important lifesaving- and damage preventive tool, which Norway must aim to implement.

In the National Vulnerability and Emergency Planning Report for 2010 (*appendix 4*), the Directorate for Civil Protection and Emergency Planning also addresses the need for establishing a more modern population warning system.

Further to the above mentioned reports and the Clim-ATIC mandate, we therefore decided to use Twitter and Facebook to convey the warning message in parallel with issuing the population warning to all fixed phones and mobile phones in the exercise area.

To our knowledge, no public offices in Norway have previously incorporated social media as a means of communicating with the general public in connection with population warning exercises

1.2 Adaptation outcome

By demonstrating a people-centred early warning system, our aim was to establish how modern technology can help reduce the negative consequences of climate change, and also help prevent the loss of lives.

A key aspect of the project has thus been to assess how a multi-hazard warning system can be established within a framework that considers the warning needs of different types of undesirable events and hazards.

In this respect, our project has aimed to demonstrate how an existing inter-municipal emergency operations centre can be a key actor in issuing population warning messages in Sogn og Fjordane county. This operations centre, *Alarmsentralen*, is a county-encompassing* monitoring and warning service that operates round the clock in order to register fire incidents and launch adequate fire fighting operations.

Alarmsentralen also helps facilitate and coordinate inter-municipal fire fighting operations, and its operations and procedures are therefore closely coordinated with relevant police and health authorities. This operational pattern ensures that Alarmsentralen responds to emergency incidents in a quick and reliable manner.

Moreover, as *Alarmsentralen* is already manned around the clock by highly qualified staff, embedding a population warning system in this organization will incur little cost, training or resources.

This fact suggests that that modern warning system technology can be combined with existing infrastructure and organizational patterns to enable local authorities to issue population warnings in a cost-effective and sustainable way.

Moreover, as climate change is generally expected to lead to more frequent natural disasters and extreme weather incidents in years to come, an early warning system will therefore constitute an important part of local communities' climate change adaptation strategies.

However, before successfully implementing a location based population warning system, confidentiality legislation and system regulations must be addressed, as discussed in more detail below.

* 24 out of 26 municipalities in the county of Sogn og Fjordane are members of Alarmsentralen. The remaining two municipalities are connected to an equivalent centre in the neighboring county of Hordaland.

Chapter 2 The warning system

In January 2005, the World Conference on Disaster Reduction adopted the "<u>Hyogo Framework for</u> <u>Action 2005-2015</u>", and emphasized the importance of developing "*early warning systems that are people centered, in particular systems whose warnings are timely and understandable to those at risk*".

An important outcome of the 2005 World Conference was the document "<u>Developing Early Warning</u> <u>Systems: A Checklist</u>", which identifies four key elements of effective early warning systems:



Further to the above mentioned documents and criteria, our project aim was to demonstrate an efficient people-centred warning system with a multi-hazard approach.

In this respect the project reference group consisted of representatives from various sectors and levels of society. As regards the "*dissemination and communication*" element above, this reference group played a key role in identifying a technical system specification aimed at enabling accurate and reliable dissemination of multi-hazard warning messages.

2.1 Project reference group

The project reference group was established in August 2008, and consisted of members from the following organizations:

- <u>The County Governor of Sogn og Fjordane PROJECT MANAGER</u> (The state's representative on regional level, being responsible for coordinating emergency planning within the county)
- <u>The Western Norway Research Institute</u> <u>Clim-ATIC regional lead partner</u> (Assignment-based research institute with competence within the field of climate research)

- <u>The Norwegian Energy Resources and Water Directorate</u> (The Directorate's mandate includes contingency planning for floods; its Sogn og Fjordane office also being the national authority on avalanches)
- <u>The National Public Roads Administration</u> (Responsible for state roads; having a special interest in population warnings related to closed tunnels and mountain crossings due to bad weather/accidents)
- <u>Telenor</u>

(Norwegian Telecommunications Company, currently the sixth largest mobile phone operator in the world)

- <u>The Norwegian Post and Telecommunications Authority</u> (Autonomous administrative agency under the Norwegian Ministry of Transport and Communications, with monitoring and regulatory responsibilities for telecommunications markets in Norway)
- <u>The Norwegian Broadcasting (in Sogn og Fjordane)</u> (The Norwegian government-owned radio and television public broadcasting company, being the largest media organization in Norway)
- <u>E-CO Hydro Power</u>

(Norwegian power company and the second largest producer of electricity in Norway. Special interest in population warnings related to dam-breaks causing floods, tsunamis)

- <u>The Norwegian Police, Sogn og Fjordane District</u> (The police is part of the Norwegian rescue service and is responsible for leading rescue work during accidents and disasters; including extreme weather events)
- <u>Alarmsentralen ("The Alarm Central in Sogn og Fjordane</u>") (Inter-municipal operations centre which facilitates and coordinates inter-municipal fire fighting operations in the county)
- <u>The Directorate for Civil Protection and Emergency Planning</u> (Maintains an overview of risk and vulnerability in Norway, and promotes measures to prevent accidents, crises and other undesirable incidents)
- <u>The Norwegian Civil Defence</u> (The State's reinforcement for emergency and rescue departments in the event of major accidents and special incidents)
- <u>Aurland Municipality</u> (Municipality in Sogn og Fjordane; having previously experienced extreme weather events and avalanches, rockslides etc. Aurland is also where the project warning exercise was held)
- <u>Unified Messaging Systems (UMS)</u> (The leading supplier of advanced message handling in Northern Europe, offering services for alerting the population in case of critical incidents)

- <u>Sogn og Fjordane University College</u> (Joined the reference group in 2009 to assist in evaluating the early warning exercise, but unfortunately had to withdraw its membership due to a lack of internal resources)
- <u>Førde Health Trust</u> (Health enterprise responsible for performing geographic and specialist activities on behalf of Western Norway Regional Health Authority)

As our angle of approach has been to demonstrate a system with a multi-hazard approach, the project reference group has played a key role in defining a system specification that enabled efficient and reliable issuing of multi-hazard warnings.

2.2 System specification

An emergency population warning is generally understood as the method by which relevant authorities can contact members of the public to warn them of an impending emergency. Hence, implementing effective and reliable early warning systems should be of high priority in areas where the frequency and scale of extreme weather is believed to increase

The overall aim of this project has been to demonstrate a modern emergency population warning by disseminating phone-based warning messages in a specified geographic area. Following discussions in the reference group, the following system criteria were considered necessary:

- The warning system must be enable the system operator to easily define the geographic reach of the warning, by the use of electronic mapping systems
- The system must be location-based so that the warning message is sent to *all* fixed phone lines and mobile phones in the chosen geographic area (i.e. just not to predefined groups / subscribers to the warning service)
- The warning must be issued both as spoken message and as a text message
- Visitors as well as permanent residents in the specific area must be alerted, and the text message must therefore be automatically translated into different languages, depending on the "nationality" of a mobile phone's SIM-card
- The warning must be given queue-priority if phone networks are busy
- The warning must be sent repeatedly until the receiver of the warning has verified that the warning is both received and understood
- There must be some communication between the receiver of the warning and the issuer, e.g. response by using the keypad:

•	The receiver must easily be able to listen to warning several times	("Press 1"?)

- The receiver must easily be able to verify that the warning is understood ("Press 2"?)
- The receiver must easily be able to notify the issuer if any assistance is needed ("Press 3"?)
- The system must register whether or not the receiver listened to the entire message

- The system must record how many people need assistance and where they are located
- The system must register whether or not the receiver of the warning has listened to the message in person, or if the warning has been received by voicemail / answering machine

As well as issuing a phone-based population warning in accordance with the system criteria, we also aimed at using Facebook and Twitter to issue the warning message during the project exercise. This concept was promoted through an article in the October 2009 issue of the Directorate for Civil Protection and Emergency Planning's quarterly magazine "Samfunnssikkerhet" (Appendix 5).

Chapter 3 Warning system suppliers

3.1 Mapping the landscape

Having defined the criteria for the phone-based population warning system, the next step was identifying potential system suppliers and forwarding the system criteria to the for feedback as regards their interest and ability to comply with the criteria and take part in the project.

Initially, three warning system suppliers responded positively to our request, whereas only two of these were still considered potential system suppliers following closer dialogue and discussions. One of these two companies was UMS – which was already a consultative member of the reference group. The second company was Fram Web, which delivers technological solutions for integrated population warning services.

Both companies were invited to attend the next reference group meeting to present their services, and they were subsequently asked to stipulate their costs for using their warning systems for the planned warning exercise in June 2010.

3.2 UMS

Being the leading supplier of advanced message handling in Northern Europe, UMS is an established actor within the field of population warning, and its systems have previously been tested at various full-scale exercises (using NetCom's infrastructure*).

UMS offers a centralized system with a GIS based user interface enabling the emergency authorities to select or draw the area on the digital map. The system will immediately identify all fix phones, cell phones and visitors within the affected area and alert them.

Furthermore, the system provides detailed logistic information, such as the number of people within the affected area, identifies types of nationality and translates the emergency messages accordingly. The system will also monitor and localize citizens responding that they are in need of help**.

* Approximately 54 % of mobile phone subscribers in Norway use Telenor's network, whereas approx. 23 % use NetCom's network. Fram Web operates on Telenor's infrastructure, as Netcom's infrastructure has been made unavailable to Fram Web due to a network availability contract which is currently being assessed by the Norwegian Competition Authority.

** As regards location-based technology, confidentiality legislation and system regulations restrict the system operator's ability to log and access this type of data.

3.3 Fram Web

Fram Web offers web-based solutions for population warning services, its Varsling 24 module ("*Warning 24*") being a set system for automatic issuing of alerts to fixed- and mobile phones.

The company's services are in part developed in response to municipalities' warning needs (e.g. in connection with contaminated drinking water), and its system has previously been used by the Norwegian Food Safety Authority to alert farmers during outbreaks of the bluetongue virus.

Telenor's integration of location-based technology into its network enabled Fram Web to offer location-based warning services using Telenor's infrastructure. The geographic reach of the warning message is defined by the system operator through the use of electronic mapping systems, and both fixed phones and mobile phones using Telenor's network (*as above) are alerted.

Like UMS, Fram Web's system offers detailed information regarding the number of people alerted etc. through the system's log module (**as above).

3.4 Choosing the warning system supplier

Both UMS and Fram Web seemed to comply with the warning system criteria, and following UMS and Fram Web's presentations to the project reference group, both suppliers were considered potential candidates for delivering phone-based warning system services for the project exercise in June 2010.

Thus, the reference group concluded that technological aspects did not seem to represent a major challenge in respect to the population warning exercise, whereas budgetary restrictions were identified as a key challenges to be overcome.

Consequently we submitted an application for project funding from the Directorate for Civil Protection and Emergency Planning (DSB), as we believed our project would be highly relevant for DSB, which has been given the assignment of mapping and testing population alert systems on a national basis. Unfortunately, our application was turned down as DSB prioritized other projects at that moment.

Having obtained price estimates from UMS and Fram Web, UMS was no longer considered a potential candidate for delivering warning system services due to its high projected costs. Fram Web's price estimate was approximately 15 % of UMS's quote, and was thus within the project's budgetary framework.

From an emergency planning perspective, it was also highly interesting to be able demonstrate an alternative to UMS, whose warning systems have previously been tested at various exercises. Subsequently, both UMS and Fram Web were informed that Fram Web was chosen as system supplier for the June 2010 population warning exercise.

4.1 Exercise context

The setting for the warning exercise was Aurland municipality, which was also a member of the project reference group. Aurland is an area of great natural beauty, but also of numerous potential hazards such as avalanches and rock slides. Aurland's somewhat challenging geography and topography makes the municipality vulnerable to extreme weather events, and thus provides an interesting setting in which to demonstrate a modern system for people-centered early warnings.



Source: <u>www.gulesider.no</u>, <u>www.fylkesatlas.no</u>

Aurland has roughly 1.700 inhabitants. The main road (E16) between Oslo and Bergen passes through Aurland, and there are several long tunnels in the area (e.g. the Lærdalstunnelen; 25 km of length, and the Gudvangtunnelen; 11 km of length).

The Flåm Railway and parts of the Oslo-Bergen railway passes through Aurland Municipality, and being a well-known cruise destination, Aurland annually attracts hundreds of thousands of tourists.



Source: <u>www.alr.no</u>

4.2 Planning

The planning of the exercise was carried out by the County Governor of Sogn og Fjordane (CG), in cooperation with the Sogn og Fjordane exercise committee. This committee supervises and coordinates large emergency and rescue exercises in the region, and consists of representatives of (the CG's Office,) the police, the Norwegian Civil Defence and volunteer organizations.

On the day of the warning exercise, the Sogn og Fjordane Police District and the CG were to carry out an "extreme weather table-top exercise" for Aurland municipality and volunteer organizations in Aurland.

So whereas the Clim-ATIC "warning message" would not relate specifically to extreme weather events, the table-top exercise would enable us to demonstrate the population warning system in a setting where local authorities and other key actors were subject to various climate-related challenges.

The purpose of the warning exercise was to demonstrate how modern technology can constitute an important part of local and regional authorities' efforts to adapt to climate change. In particular, our intention was to send out a phone-based warning message in Aurland municipality to determine the reliability and efficiency of such systems.

We also decided to use Twitter and Facebook to convey the warning message in parallel with issuing the population warning to all fixed phones and mobile phones in the exercise area.

The population warning message would thus be sent to both fixed phones and mobile phones in Aurland, as well as being posted on Facebook and Twitter

<u>Facebook</u> is a social networking website, with approx. 2.57 million users in Norway. On June 1st 2010, the CG's Facebook-profile had 260 followers, who automatically received news published on the CG's website. These 260 followers could then share the news with their friends and contacts on Facebook, and large numbers of people could thus be reached in a short space of time.

<u>Twitter</u> is a social networking and micro-blogging service that enables the users to send out and read other user messages called tweets. Twitter has approximately 130.000 users in Norway, and during the warning exercise, the warning message was also published on Twitter.

In addition to obtaining first-hand knowledge on how local authorities would respond to a phonebased population warning system, we also aimed to assess whether the use of such systems could improve the interaction between rescue organizations, local authorities and volunteer organizations.

As regards the evaluation of the warning exercise, Sogn og Fjordane University College originally intended to assist in evaluation process, but had to withdraw from the project due to a lack of internal resources. We therefore decided to combine the warning system's log with the use online evaluation forms and a limited door-to-door survey in parts of Aurland (Høydalen) to evaluate the Clim-ATIC warning exercise.

4.3 Dissemination

Before carrying out the population warning exercise, several dissemination activities were undertaken by the CG's Office. Information about the exercise was published on the CG's website, Twitter profile and Facebook account, as well as being promoted through the local newspaper (*Sogn Avis*) and the county-wide district radio station (*NRK Sogn og Fjordane*).

In order to reach as many people as possible in the exercise area, the phone warning would combine population-based technology and location-based technology. The population-based technology would reach all fixed phones (*voice message*) and mobile phones (*text message*) registered in Aurland municipality, whereas the latter would reach all mobile phones using Telenor's infrastructure (*text message*) in Aurland at the time of the exercise.



The County Governor's website; prior to the Clim-ATIC warning exercise

4.4 The demonstration exercise

The Clim-ATIC warning exercise was held in parallel with a table-top exercise focusing on local authorities' ability to respond to extreme weather events. Several of the reference group members attended the exercise as observers.

According to the table-top scenario, the Norwegian Geotechnical Institute (NGI) had recently issued a general avalanche warning for Sogn og Fjordane county, and the Norwegian Energy Resources and Water Directorate had also issued a flood warning for the same area. Several roads in the county had already been blocked by avalanches and landslides, and the National Public Roads Administration accordingly warned motorists to be on the alert due to the severe weather conditions.

As Aurland municipality's crisis management team was discussing relevant measures in response to the exercise scenario, Fram Web issued the Clim-ATIC warning message to demonstrate how local authorities can use modern technology to convey important information to the public under challenging circumstances. To supplement the phone-based warning message, Twitter and Facebook were used to send out information about the exercise.



Members of Aurland municipality's crisis management group



"EXERCISE – The County Governor and Aurland municipality are testing a population warning system. You receive this message because you are currently in the exercise area. More info on <u>www.fmsf.no</u>" (The County Governor's website)

In general, phone-based warning messages can be issued by the use of two different technological platforms; in this report labeled *population-based technology* and *location-based technology*.

Using *population-based technology*, Aurland residents would receive the phone-based warning message as voice message to fixed phones* and as text message to their (Telenor-based) mobile phones. Hence, residents listed with both fixed phones and mobile phones would receive both types of warnings. Accordingly, mobile phones "residing" in Aurland but not being in Aurland during the exercise, would also receive the warning message. Tourists and others travelling through Aurland would not receive this warning message.

Using *location-based technology*, all (Telenor-based) mobile phones within the geographic area defined by the system operator would receive the warning as a text message. Hence tourists "roaming" on Telenor's network would also receive the warning.

Both fixed phones and mobile phones in Aurland thus received the population message during the Clim-ATIC warning exercise. Some 2,500 mobile phones using Telenor's network received the population alert as text message, whereas 322* fixed phones in Aurland received the population alert as voice message.

* This number of fixed phone lines in Aurland municipality was generated by a search engine using two principles:

- In densely populated / urban areas the system identifies the street addresses and its registered residents' home phone number accordingly.

- In rural areas with "informal" addresses (i.e. no street names), the system identifies the property address, and looks up the registered owner of the specific property. Public offices etc. are often not the registered owner of their respective office-buildings, and this latter method therefore leads to a somewhat lower result-percentage than the method based on street address.

To compensate for this gap, Fram Web has developed a system where public offices, schools, kindergardens etc. can register their "resident" address to be included in the "property-search".

Following the voice message, the receiver was given two alternatives for response: "*Press one to verify that you have understood this message*" or "*press nine to listen to the message again*". Consequently, Fram Web's logging system featured detailed status and statistics for this population-based voice message, and the system instantly registered who and how many who had listened to, or listened to and verified, the voice message*.

As regards the location-based warning message, this technology identifies where specific mobiles are positioned at specific times, and is thus subject to confidentiality legislation. As a result, Fram Web's current logging system does not provide data from the location-based warning. However, Telenor could inform us that some 2,500 messages had been sent out.

Ideally, we wanted to make our Facebook-advertisement visible only to Facebook users based in the county of Sogn og Fjordane, but the only "regional label" available was "Norway". Our exercise-advertisement could therefore be seen by all Norwegians above the age of 13 years, which is the lowest age limit to register on Facebook.

The advertisement was scheduled to be "active" (*i.e. visible to Facebook users in Norway*) between 10 a.m. – 1 p.m. on June 2^{nd} 2010. However, as Facebook must manually approve all "advertisements" that are to be published; our warning exercise advertisement was not made visible until 11 am on June 2^{nd} .

Thus, the warning exercise advertisement was only visible for two hours, during which it had 201 849 viewings, and was clicked on 90 times. Moreover, the CG's Facebook profile got 15 new followers this day, and the largest increase of followers was in the age-group 13-17 years. During the warning exercise, the warning message published on Twitter was re-tweeted (*"shared/forwarded"*) by 6 or 7 persons.



Information published on Twitter (left), and Facebook (right)

* When issuing "group warnings" to predefined groups/phone lists, the system operator can easily add multiple-response alternatives, such as "press two if you need assistance". This feature is currently not available when issuing population-based warnings, but would be highly important in a real crisis situation, as the logging system would then provide the system operator with a complete overview of the situation, thus enabling him to e.g. direct emergency personnel to those in need. Both the voice message and the text message referred people to the CG's website to obtain more information on the warning exercise. Those who visited the website were encouraged to answer a few questions regarding how they received the warning message, by filling in an online evaluation form.

Additionally, information about the warning exercise and the electronic evaluation form was published using the County Governor's Facebook- and Twitter-profiles, and people were encouraged to share this information with others.

Clim-atic - Varslingso Fylkesmannen i Sogn og Fjo	aving 2010 Irdane	0%	50% 100%
Early Warning Execise 2	010		Help
Compulsory questions are mark	ked with an asterisk*		
Gender:*	🖾 Male 💿 Female		
Age:*	•		
Lives in Aurland:®	💿 Yes 💿 No		
Visiting Aurland:*	🔘 Yes 🔘 No		
How did you receive the early warning alert? *	 Fixed Phone line Mobile phone Radio / tv Paper / online newspaper Facebook Twitter Other: 		
Was the alert easy to understand?*	🔘 Yes 🔘 No 🔘 I don't know		
Post a comment on the warning	g exercise (optional field)	_	
Please press "next page" to ver	rify and submit you answers		

The online evaluation form, which was completed by 88 persons

Chapter 5 Exercise evaluation

By combining Fram Web's logging system, an electronic evaluation form and a door-to-door survey, we aimed to measure the efficiency of the warning system by determining the following:

- Did the warning message reach everyone in the exercise area?
- How many were alerted?
- Were the community's most vulnerable reached?
- How were people alerted? Phone or other means of information?
- Did the recipients understand the message?

5.1 Fram Web's logging system

The voice message was sent out at 10.30 a.m., with a repetition call one hour later. Following the voice message, the recipient was given two alternatives for response: "*Press one to verify that you have understood this message*" or "*press nine to listen to the message again*".

Consequently, Fram Web's logging system featured detailed status and statistics for this populationbased voice message, and the system instantly registered who and how many had listened to, or listened to and verified, the voice message. The logging system registered that 156 out of the 322 phone calls were answered, whereas 71 of the calls were both answered and verified.

This leaves 95 unaswered phone calls. Some of these 95 people might not have been at home during the exercise, and we also know that some of these 95 persons received the warning as text message to their mobile phones. However, the number of unanswered calls might have been reduced if the number of repetition calls had been increased.

					I varining	1
🔶 Beredskap	svarsling	Denne - Tale	e siden viser en detalje '	oversikt over for var	slingen 'Aurland	Svein Helge
En tjeneste i Varsling	24	Det - Ta	aljer for befolk ale'	ningsvarsling '/	Aurland	Steinsåker profil logg ut
		\$ 1	ilbake til oversikt 📢	Oppdater siden		Sogn og Fjordane
Statuslogg			1	Statur	19	Hielo 🥠
30.05.2010 09:55:25				Planlagt		
MS_cratictikk						
Туре	Antall Leve	rt til mobil	Sendt til opera	ıtør, ikke levert	Feil	Contraction of the local division of the loc
Adressebasert	0	0 (0 %)		0	0	600
Fasttelefon (talemeldi	ng) statistikk					
Totalt antall	Ubesvart	Besvart	Besvart og	bekreftet	Feil	
322	0 (0 %)	0 (0 %)		0 (0 %)	0	To and
Personer inkludert i b	efolkningsvarsli	ng				and the second
Hilstad, Terje 30.05.2010 09:55:26	57633494	Ikke ringt				
Vidme, Steinar 30.05.2010 09:55:26	57633955	Ikke ringt				
Bjelde, Olav 30.05.2010 09:55:26	57631187	lkke ringt				
Bjelde, Sverre 30.05.2010 09:55:26	57631117	Ikke ringt				
Bjørgum, Terje 30.05.2010 09:55:26	57633008	lkke ringt				
Hjellum, Nils 30.05.2010 09:55:26	57631770	ikke ringt				
Hjellum, Sten 30.05.2010 09:55:26	57631718	lkke ringt				
Sjursen, Harald						

Fram Web's population-warning logging system

No errors were registered in respect to issuing the voice message. Location-based technology identifies where specific mobiles are positioned at specific times, and is thus subject to strict confidentiality legislation. Subsequently, Telenor did not provide statistics regarding the location-based text message, and Fram Web's logging system did provide data from the location-based warning. However, Telenor informed us that some 2,500 text messages had been sent out.

Whereas the text message was sent out at 10.30 a.m., it took roughly 5-10 minutes before the majority present at the table-top exercise had received the text message. Moreover, it took up to an hour before everyone present at the table-top exercise had received the warning message.

Unfortunately, we do not hold any information regarding how this location-based warning affected the local phone network, or if local network limitations / restrictions were the cause of the delay.

Telenor stated that the warning message was being continuously processed and sent, and that all messages were ultimately successfully delivered (even though it took a bit longer than we had anticipated).



Fram Web's general manager, Mr. Jarle Heimtun, demonstrates the warning system to representatives from the Sogn og Fjordane Alarm Central and the police.

5.2 The online evaluation form

Whereas Fram Web's logging system provided us with important information about the phone warning, the online evaluation form and the door-to door survey also enabled us to determine how successful and efficient the use of Facebook and Twitter had been.

The warning exercise advertisement which was published on Facebook was visible for two hours, during which it had 201.849 viewings, and was clicked on 90 times. Data gathered from the CG's Facebook-profile tells us that CG got 15 new followers on Facebook during the warning exercise, six of whom were aged between 13-17 years. This latter fact suggests that social media is an important arena for reaching young people with important information. The warning message that was published on Twitter was re-tweeted ("*shared/forwarded*") by 6 or 7 persons, which is a somewhat lower figure than we had anticipated.

The online evaluation form was completed by 88 persons, several of whom had received the warning message in more than one way (*appendix 6*). Importantly, 70 per cent of those who filled in the online evaluation form had received the warning message as text message to their mobile phones.

This illustrates that the use of location-based warning system is by far the most efficient method of alerting the general public during crisis situations. Subsequently, this underlines the fact that issues concerning confidentiality legislation and system regulations must be addressed so that efficient and sustainable population warning system can be developed and implemented

Furthermore, 10 per cent of the respondents received the warning via Facebook, and this figure suggests that the use of social media in some cases can be as efficient as a "traditional" population-based phone-warning (voice message).

Korleis fekk du varsele	et?			
Alternativ		Tal svar		Svar-del
Fast telefon		9		8,4 %
Mobiltelefon		78		72,9 %
Radio/tv		2		1,9 %
Avis/nettaviser		0		0,0 %
Facebook		11		10,3 %
Twitter		3		2,8 %
Anna		4		3,7 %
80 70 60 50				
30				****
20				The second s
10				
0 Fast telefon Mobil	tel Radio/tv A	vis/netta Facebook	Twitter A	nna

From the online evaluation form: "How did you receive the warning message"?

Some of the 88 respondents stated that they had seen the warning information online, whereas they had not received the warning message to their mobiles or through other channels. This emphasizes the importance of using supplementary warning systems.

The use of radio has previously proved to be an efficient method of conveying various types of messages to the general public, and in a real crisis-situation the regional radio station (NRK Sogn og Fjordane) would probably have been one of the most important ways of providing (supplementary) information to the general public.

Additionally, during a real crisis both local and regional media would doubtlessly have responded to any tweets being sent to them, thus leading to wide media coverage of the actual event.



From the online evaluation form: "Was the warning message easy to understand"?

The majority of the respondents stated that the warning message was easy to understand. Of the 15 per cent who answered "no" or "don't know", several people commented that the message was understandable, whereas they did not understand why it was being sent out.

In a real crisis-situation, such warning messages would probably identify the type of hazard (e.g. *"high risk of avalanches in Aurland"*) and instruct the public on further action (e.g. *"evacuate the area"*) – thus clearly illustrating to the public *why* the message was being issued.

5.3 The door-to-door survey

In order to obtain a more detailed understanding of how the general public in Aurland reacted to the warning message, Aurland Red Cross carried out a door-to-door survey in the "Høydalen" area of Aurland municipality.



Left: Høydalen, Aurland; source: <u>www.fylkesatlas.no</u> <i>Right: Aurland Red Cross planning the door-to-door survey

In total, 31 people were interviewed, but not everyone answered all of the questions (appendix 7). Approximately 77 per cent received the warning message to their mobile phones, whereas 17 per cent received the warning to their fixed phone lines.

The vast majority said the message was easy to understand.



From the survey: "How did you receive the warning" (left); "Was it easy to understand" (right)?

The survey also revealed that not all family members of the households in Høydalen had received the warning message. This could be a result of some people not owning a mobile phone, or not having had access to Facebook etc. However, 75 per cent of the people in Høydalen responded that everyone in the family had received the message, again emphasizing the efficiency of the warning system.

As discussed above, incorporating both population-technology and location-technology in logging systems would enable to operator to identify households without response, so that emergency personnel etc. could be directed to these addresses.



From the survey: "Did everyone above the age of 12 receive the warning message"?

Importantly, the survey conducted in Høydalen revealed that 100 per cent of the respondents wanted to receive population warning messages to their phones and through social media during extreme weather events and other challenging situations.



From the survey: "Do you wish to receive warning messages to your phone and through social media"?

Chapter 6 Project summary and concluding remarks

The objective of the *Emergency Population Warning System* project has been to demonstrate an effective, reliable and cost-efficient early warning system with a multi-hazard approach. This report has illustrated how a phone-based warning system can be combined with the use of social media in order to convey important information to the general public.

6.1 The warning system

- We have argued that the technical aspects of people-centered warning systems are at large readily available, whereas issues concerning confidentiality legislation and system regulations must be addressed before successfully implementing efficient location-based warning systems.
- As we have demonstrated, publishing advertisements using social media is also subject to regulations. In order to issue warnings which are *"timely and understandable to those at risk"*, we believe more research needs to be carried out on the opportunities and restrictions connected with the use of social media during crisis situations.

6.2 The system operator

- In peacetime, the task of deciding *when* a population warning is to be issued in Norway lies with the police. The police also has the responsibility for carrying out adequate measures to evacuate the public when needed. However, the police has no formal responsibility for developing population warning systems.
- In order to establish a cost-effective early warning system, multi-hazard approaches are a
 prerequisite, as the costs of using and maintaining the system will be shared. A locationbased warning system can also represent an integrated tool for disseminating purely
 informative messages, such as general public information or various messages from local
 authorities.
- Hence, this project demonstrated how an existing county-encompassing organization could be used to issue the population warning. As this organization is closely coordinated with the police whilst being an inter-municipal organization, it is thus suited for the issuing of both non-emergency and emergency warnings with a multi-hazard approach.

6.3 Project relevance and follow-ups

"Countries that develop... institutional frameworks for disaster risk reduction...have greater capacity to manage risks and to achieve widespread consensus for, engagement in, and compliance with disaster risk reduction measures across all sectors of society"

Hyogo Framework for Action 2005-2015: Building the Resilience of Nations and Communities to Disasters, Paragraph 16

Several communities in both Sogn og Fjordane and the wider NPP-area are facing numerous hazards such as flooding, avalanches, rock slides and other extreme weather events. Great distances between peripheral communities and at times challenging infrastructure can make accessibility poor, thus making communities more vulnerable to extreme weather events.

By implementing emergency population warning systems, local and regional authorities can effectively inform and communicate with the general public, and such systems can also improve rescue operations etc. by diverting and optimizing available resources.

Implementing effective and reliable early warning systems should therefore be of high priority in areas where the frequency and scale of extreme weather is believed to increase.

The *Emergency Population Warning System* project has been mentioned in a proposition report by the Norwegian National Defence Committee, which discussed the need for a modern system for population warning. The majority of the committee stated that efficient population warning by the use of modern technology constitutes an important lifesaving and damage preventive tool, which Norway must aim to implement.

Moreover, in the National Vulnerability and Emergency Planning Report for 2010, the Directorate for Civil Protection and Emergency Planning also addresses the need for establishing a more modern population warning system. The directorate subsequently recommended that a project group consisting of representatives from the directorate and the police is set up to assess the possibility of implementing a nation-wide phone-based warning system.

The EU is currently assessing and testing Cell Broadcast *, which is a phone-based technology for communicating with the general public in high-risk and crisis situations. Furthermore, The European Emergency Number Association (EENA) already advocates harmonized emergency telecommunications in accordance with European requirements.

Based on the experiences we have drawn from our project, we support the Directorate for Civil Protection and Emergency Planning's initiative to set up a national reference group to assess the development and implementation of a country-encompassing people-centered warning system.

As we have argued in this report, technological aspects seem to represent less of a challenge than issues concerning confidentiality legislation and system regulations. Hence, we recommend that the reference group's mandate incorporates the following:

- Confidentiality legislation and system regulations
- Opportunities and restrictions connected with the use of social media during crisis situations
- Transnational aspects of population warning, as illustrated by the EU's Cell Broadcast initiative

* https://service.projectplace.com/pub/english.cgi/0/283748154

Appendix

Appendix 1	Examples of adaptation measures	p. 26
Appendix 2	Project plan	p. 27
Appendix 3	National Defence Committee Proposition Report Innst. S. nr. 85 (2008-2009)	p. 29
Appendix 4	National Vulnerability and Emergency Planning Report, 2010	p. 30
Appendix 5	Samfunnssikkerhet (3, 2009) - the Directorate for Civil Protection and Emergency Planning's quarterly magazine	p. 31
Appendix 6	Statistics from the online evaluation form	p. 32
Appendix 7	Statistics from the survey	p. 33

Appendix 1 Examples of adaptation measures

EUROPE		
The Netherlands, Government of the Netherlands (1997 and 2005)	Sea-level rise	Adoption of Flooding Defence Act and Coastal Defence Policy as precautionary approaches allowing for the incorporation of emerging trends in climate; building of a storm surge barrier taking a 50 cm sea-level rise into account; use of sand supplements added to coastal areas; improved management of water levels through dredging, widening of river banks, allowing rivers to expand into side channels and wetland areas; deployment of water storage and retention areas; conduct of regular (every 5 years) reviews of safety characteristics of all protecting infrastructure (dykes, etc.); preparation of risk assessments of flooding and coastal damage influencing spatial planning and engineering projects in the coastal zone, identifying areas for potential (land inward) reinforcement of dunes.
Austria, France, Switzerland Austrian Federal Govt. (2006); Direction du Tourisme (2002); Swiss Confederation (2005)	Upward shift of natural snow- reliability line; glacier melt	Artificial snow-making; grooming of ski slopes; moving ski areas to higher altitudes and glaciers; use of white plastic sheets as protection against glacier melt; diversification of tourism revenues (e.g., all-year tourism).
	Permafrost melt; debris flows	Erection of protection dams in Pontresina (Switzerland) against avalanches and increased magnitude of potential debris flows stemming from permafrost thawing.
United Kingdom Defra (2006)	Floods; sea-level rise	Coastal realignment under the Essex Wildlife Trust, converting over 84 ha of arable farmland into salt marsh and grassland to provide sustainable sea defences; maintenance and operation of the Thames Barrier through the Thames Estuary 2100 project that addresses flooding linked to the impacts of climate change; provision of guidance to policy makers, chief executives, and parliament on climate change and the insurance sector (developed by the Association of British Insurers).

Source: IPCC raport 4:

http://www.ipcc.ch/publications_and_data/ar4/wg2/en/ch17s17-2-2.html#table-17-1

Appendix 2 Project plan



Helge Asperheim, (Mr.), The Directorate for Civil Protection and Emergency Planning (DSB) Steering group meetings will take place at least every six months, or more often if required. The first meeting will be held before November 1st 2008. 4.2. Project Manager The local project manager is Haavard Stensvand, Head of the County Governor's Emergency Management Office. The local project management arrangements The local project manager will be responsible for establishing sufficient procedures for: 4.3. Project management arrangement Monitoring and controlling project progress and costs (and, if necessary, proposed recovery actions) · Monitoring and controlling quality Assess validity of project objective statement (and if necessary, propose changes) Controlling and resolving issues that arise during the process · Reporting to regional steering group and WP4-leader 5. Project Activities and Deliverables

Activities to be undertaken 5.1.

See enclosed appendix

5.2. Deliverables

A comprehensive report describing relevant experiences will be compiled within the end of the project period.

Clim-ATIC WP4 Generic Project Plan - Draft - May 2008

 Carlo Aall (Mr.) Idun A. Husabo (Ms.), Western Norway Research Institute (WNRI) Anne Karin Hamre (Ms.), County Governor of Sogn og Fjordane (SFCG)

5.3. Social and/or economic benefits

The final report will shed light on benefits of various kinds, including:

Adaptation of technology - The project entails a possibility for adaptation of existing technology to the needs of local communities in Norway.

Page 5 of 10

Clim-ATIC WP4 Generic Project Plan - Draft - May 2008

The results of the project (in particular the full scale test) may be a useful basis for the discussions around the potential establishment of a permanent system.

5.6. Closing out the project After the project is completed, all project related material (project data) will be archived by the County Governor's office.

As a part of the completion process, the County Governor will conduct a session with the local project team members, with the objective of putting emphasis on the lessons-learned.

6. Project Reporting

6.1. Reports to WP4 leader

Progress reports will be compiled and sent to WP4 leader every three months -the first mid December 2008.

The reports will comprise information about status (schedule and budget), accomplishments, next steps, predicted slippages etc.

6.2. Meetings with WP4 leader Meetings (by telephone or personal) with WP4 leader will be held at least every six months. The first meeting was held in Bergen in the beginning of October 2008.

Needs of further meetings will be continuously considered and discussed with WP4 leader through the whole process.

6.3. Key project milestone A brief stage end report will be compiled and sent WP4 leader before the final preparations for the full scale exercise start up.

The report will create a basis for the undertaking of a comprehensive mid project review, and will comprise information about Summary of work carried out so far

Page 7 of 10

Clim-ATIC WP4 Generic Project Plan - Draft - May 2008

- Improved disaster management Lessons from this demonstration project may benefit the County Governors of Norway as well as other actors within civil protection and emergency planning, in the sense that new tools and forms of communication will hopefully be tested and improved on their behalf.
- Better coordination The implementation of an early warning system would most likely serve to clarify responsibilities and improve cooperation across organisations and institutions in the context of disaster management.
- Identify need for training knowledge The demonstration project will identify training or knowledge-related gaps in disaster management, thus indirectly contributing to enhancing learning and raising awareness.

Transnational aspects 5.4.

Reliable systems of public early warning may in different ways represent important and necessary parts of adaption strategies for communities and regions exposed by increased threats of climate-related hazards.

- In exposed by increased tureats of cumate-retares mazaros. A common aim for most regions represented in the project, is econo development through increased to unian. For the tourism industry, pu safety is an important competitive advantage. Increasing number natural hazards may represent tough obstacles in the effort to bring in tourists into the relatively peripheral Nordic regions. stry, public
- vuunts moo the relatively pempheral Nordic regions. The Clim-Aide project a civities in Faland includes a GIS-based flood nik management system for the rivers Kemijoki and Osnatjoki in Rovanema (developing flood nisk management plans, visualizations for local people, building codes and land use plans). As early varning system may represent an important integral part of the flood nisk management plans. .
- Use of mobile phone Bluetoothtechnology in the tourism demonstration project proposed for the Cairngorms National Park, Scotland will provide an important comparison for this project.

5.5. Potential continuation of the project There may be a possibility that a demonstrated early warning system may be established on permanent basis after the completion of the Clim-ATIC project.

For example, if the Rock Slide Project (see "Links to other projects" for more detail)identifies a mountain indge where there is active movement that may represent a threat of a large rock lide, an early warning system might be identified as a necessary precaution to meet the nik.

Page 6 of 10

Clim-ATIC WP4 Generic Project Plan - Draft - May 2008

• Tasks necessary to carry out in the next stages of the project Assessments about needs of prospective revisions of approach, schedule or budget

7. Dissemination

- Send out press release to regional and local media describing the demonstration project (after kick-off meeting)
- Informative article at www.fmsf.no and other relevant websites (after the kick-off meeting)
 Organise press conference and media observation at testing session
- (during testing stage)
- Recommend further action or testing (after evaluation reporting stage) Submit article in thematic magazine, e.g. "Samfunnssikkerhet", "Plan", "Cicerone" (after reporting stage)

8. Links to other projects

Other Clim-ATIC demonstration projects:

Other Clim-ATEC demonstration projects: The project is liked to other demonstration projects in Clim-ATIC: A GIS flood fink management system to be tried out in Kittilä and in Rovaniemi (the ivers Kenigioki and Ounayolo). Finland in a dätinto the Caimgeomy project, mentioned above, a river basin planning project is to be carried out in Glen Uguhant, Scolland with one objective bengt to adapt to an increasing rink of flooding. An early warning system maxy, in the near future, become an integrated part of Bood inak management plans, and early warning systems may be tested in Fuland and Scotland.

Local related projects:

The Rock Slide Project: An ongoing project in the county of Sogn og Fjordane is currently attempting to identify mountains ridges where there is active movement that may ergenent a threat of a large rock slide and related tsunamis. The County Governot of Sogn og Fjordane is a project partner.

The Natural Hazards Project (KS Naturskade): A project camied out by Western Norway Research Institute in 2007-8 for the Norwegian Association of Local and Regional Authonities. The objective of the project it to examine the ways in which a set of Norwegian communities have responded to natural

Page 8 of 10

	Chivi, ATIC WP4 Gamaric Projec	then Dirett	38 200	100			Chas ATIC WRIGH	many Propert Pl	ALC - Dec & Street 200	0.0	
	competition of a contract troper	s man - man	- May 200	18			Cull-Aric wird of	and a repetition	an - Dran - May 200	US	
disasters, to climat	, and looked at ways of mitigatin te change.	ig the risk of na	ıtural disast	ters related			TOTAL		2,927.60 3,513.00	3.513.09	0.00
Civilelin internatio in Norwa protectio	m (Civil Protection and Clim onal project on climate vulnerabi ay, Sweden and the Netherlan on and climate vulnerability.	ate Vulnerabi lity and civil pr ds, looks at va	lity): This otection, wi arious aspe	three-year ith partners ects of civil							
Noradaş Norweğis developr strategies Westem	pt (Community Adaptation and an municipalities take part in this ment of local vulnerability studies s. The municipality of Flora take Norway Research Institute.	Vulnerability sproject, which s as well as corr s part in this pro	in Norway involves th imunity ada oject togeth	r): Eight he aptation her with							
Klimatil klimatilp conseque Norway Fjordane	Ipasning no: An official web por basing no will offer advice and i ences of climate change and v Research Institute, together wit e, have contributed to a county 'p	ttal dedicated t information or ways of adapt th the County ilot' forthis por	o climate a a e.g. the va ing to then Governor o rtal by colle	adaptation, arious local m. Western of Sogn og ecting local							
Erma (E sized cor The mai tailored t them in manager 9.	Electronic Risk Management A mmunities). in objective of the ERMA-pro to the needs of small and medium the creation and efficient m ment tasks they are in charge of ment fasts they are in charge of	rchitecture for oject is to dev n-sized comma anagement of f (http://erma.g	r small and elop suppo nities, in ord ? the indivi project org	d medium- ort-systems der to assist idual crisis).							
Erma (E sized cor The mai tailored t them in manager 9.	Electronic Risk Management A mmunities). in objective of the ERMA-pro to the needs of small and medhum the creation and efficient m ment tasks they are in charge of Project Finance 1. Suffing cosh industing cost	rchitecture fo oject is to dev n-sized comma anagement of f (http://emma-p	r small and elop suppo nities, in ord ? the indivi project org	d medium- ort-systems derto assist idual crisis).							
Erma (E sized cor The mai tailored t them in manager 9.	Electronic Risk Management A mmunifies). in objective of the ERMA-pro to the needs of small and medium the creation and efficient m ment tasks they are in charge of Project Finance	rchitecture fo oject is to dev n-sized comma anagement of f (http://enma-p	r small and elop suppo nities, in ord ? the indivi project org cost 2009	d medium- ort-systems derto assist idual crisis).	cost 2011	Tetalcost 0/10/1					
Erma (E sized cor The mai tailored t them in manager 9. 9. 9.	Electronic Risk Management A mmunifies). in objective of the ERMA-pro to the needs of small and melsum the creation and efficient m ment tasks they are in charge of Project Finance 1.5 mfmg con including social county governor	rchitecture for oject is to dev n-sized comma anagement of f (http://erma-g cost 2008 27997.4	r small and elop suppo nities, in ord ? the indivi project org cost 2009 3359?	d medium- ort-systems derto assist idual crisis). <u>cost 2010</u> 33597	cost 2011 0	Total cost 95191.4					
Erma (E sized cor The mai tailored t them in manager 9. 1 work package WP 4	Electronic Risk Management A munufies). in objective of the ERMA-pro to the needs of small and medbum the creation and efficient m ment tasks they are in charge of Project Finance 5.5mf/mg.onnincleling.social county governer TOTAL	cont 2008 cont 2008 cont 2008 27997.4	r small and elop suppo nuties, in ord ? the indivi project org 33597 33,597.60	d medium- ort-systems der to assist idual crisis).	cost 2011 0 0.00	Total cost 951914 95,19149					
Erma (E sized cor The mai tailored t them in manager 9.	Electronic Risk Management A munifies). in objective of the ERMA-pro to the needs of small and medium the creation and efficient m ment tasks they are in charge of Project Finance 5 m/mg cosh including social county governer TOTAL	rchitecture for oject is to dev n-sized comma anagement of f (http://emma-p cost 2008 27997.4 27,997.40	r small and elop suppo uities, in ord the indivi project org 33597 33,697.60	d medium- ort-systems der to assist idual crisis).	cost 2011 0	Total corr 951914 95,191.40					
Erma (E sized cor The mai tailored t them in manager 9.	Lectronic Risk Management A munufies). in objective of the ERMA-pro- to the needs of mull and medium the creation and efficient m ment tasks they are in charge of Project Finance 1. Staffing costs including social county governer TOTAL 5. Local officerons (distributed by fin met)	rchitecture for oject is to dev n-sized comma anagement of f (http://emma-g cost 2008 27997.4 27,997.40	r small and elop suppo- nities, in ord the indivi- project org 33597 33,597.00 cost 2009	d medium- ort-systems der to assist idual crisis). cost 2010 33,597.00 cost 2010	cost 2011 0 0.00 cost 2011	Testal cost 95191.4 95,191.40 Total cost					
Erma (E sized con The mai tailoredt them in manager 9.	Lectronic Risk Management A munifies). in objective of the ERMA-pro to the needs of small and medium the creation and efficient m ment tasks they are in charge of Project Finance 1. Suffrag cosh including social costby devision county governor TOTAL 5. Local office som (distributed by file 1996) premises costs	continecture for opect is to dev n-sized comma- nanagement of f (http://enma-p cont 2008 27997.40 27,997.40 it cont 2008 679	r small and elop suppo nities, in ord? ? the indivi project.org? 33,597.00 cost 2009 815	d medium- ort-systems der to assist idual crisis). cost 2010 33,597.00 cost 2010 815	cost 2011 0 0.00 cost 2011 0	Tetralcost 931914 95,19140 Tetralcost 2,309					
Erma (E Sized con The mai tailored t them in manager 9.	Electronic Risk Management A munifies). in objective of the ERMA-pro to the needs of small and medium the creation and efficient m ment tasks they are in charge of Project Finance 1. Suffing cost includingsocial county governer TOTAL 5. Local efficience (distributed by fla meth) premises costs Talaphone	cost 2008 cost 2	r small and elop suppo- nities, in ord 7 the indivi- 7 the indin- 1 the indin- 1 the indivi- 1 the i	d medium- ort-systems derto assist doual crisis). cost 2010 33597 33,597.00 cost 2010 815 220	cost 2011 0 0.00 cost 2011 0 0	Total cost 951914 9519149 Total cost 2,309 633					
Erma (E sized cor The mai tailored t them in manager 9.	Electronic Risk Management A munifies). in objective of the ERMA-pro to the needs of small and medium the creation and efficient m ment tasks they are in charge of Project Finance 1. Suffrag cosh including social cost by sevence TOTAL 5. Local efficiency (distributed by file 1990) For the sevence Total Fortage	rchitecture fo oject is to dev neized comman anagement of d f (http://ema.g 0977.4 27,997.4 27,997.4 1 cost 2008 5 5 9 1 8 1 8 1 1 8	r small and elop suppo suppo totes, in ord to the indivi- totes org 33597 33,597.00 815 815 8220 220 220	d medium- ort-systems der to assist dual crisis (). cost 2010 33,597.00 cost 2010 815 220 220	cost 2011 0 0.00 cost 2011 0 0 0 0	Totalcost 53191.4 55,191.40 Totalcost 2,300 62					
Erma (E sized cor The mai tailored t them in manager 9. 1 work package WP 4 WP 4 WP 4 WP 4 WP 4	Electronic Risk Management A munifies). in objective of the ERMA-pro to the needs of small and medium the creation and efficient m ment tasks they are in charge of Project Finance 1. Smfling cosh includingsocial county governe County governe TOTAL 5. Local efforces to (distributed by fin rets) premises costs Totapa Fostapa Photocopying	rchitecture fo oject is to dev nized command anagement of f f (http://erma-g cost 2005 27/9774 27,99740 it cost 2005 579 18 219 18 219 219	r small and elop suppo- nities, in ord the indivision project org 33597 33,697.00 eost 2009 815 220 222 30	d medium- ort-systems der to assist dev to assist double control 33597 	cost 2011 0 0 0 0 0 0 0 0 0 0 0 0	Total cont 951914 9519240 Total cont 2,300 62 85					
Erma (E sized cor The mai tailored t them in manager 9.	Clectronic Risk Management A munifies). in objective of the ERMA-pro to the needs of small and medium the creation and efficient m ment tasks they are in charge of Project Finance Suffrag cosh including/social coastly govenor TOTAL E. Local office som (distributed by file mb) premises costs Talaphone Prostage Prostage Prostage Prostage Office Supplies	schitecture fo oject is to dev n-sized comma anagement of (http://erma-g cost 2008 27997.4 27,997.4 37,997.4 16 679 183 18 135	r small and elop suppo utiles, in order f the indivi- project.org 33597 33,697.60 eost 2009 815 220 22 30 0 1,627	d medium- ort-systems derta assist- idual crisis). cost 2010 33,597.00 cost 2010 815 220 20 90 1,627	cost 2011 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	Tetral cost 95191.4 95,191.40 Tetral cost 2,200 423 462 460					
Erma (E sized cor The mai tailored t them in manager 9. 9. 9. 9. 9. 9. 9. 9. 9. 9. 9. 9. 9.	Clectronic Risk Management A munufiels). in objective of the ERMA-pro to the needs of small and meshur the creation and efficient m ment tasks they are deficient and county governe county governe TOTAL 5. Loci efforces (dismbumed by fit mb) premises costs 1deproce Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Postage Posta	rchitecture fo oject is to dev nized comman anagement of f (http://erma-g cost 2008 27997.4 27,997.4 t cost 2008 153 153 153 153 153 154 155 155 155 155 155 155 155 155 155	r small and elop supponities, in order ? the indivision of the indivision project org? 33597 33597.00 815 220 220 220 220 220 220 220 220 220 230 220 230 220 230 23	d medium- ort-systems idual crisis). cost 2010 33597 33,597.00 815 220 30 222 30 31,697.00	cost 2011 0 0.00 cost 2011 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	Total cost 951914 951914 7otal cost 2,00 40 40 83 4,60 1,642					
Erma (E sized cor The mai tailored t them in manager 9. 9. 9. 9. 9. 9. 9. 9. 9. 9. 9. 9. 9.	Clectronic Risk Management A munufies). in objective of the ERMA-pro- to the needs of small and medium the creation and efficient m ment tasks they are in charge of Project Finance 1. Sinfing costs including social county governe TOTAL 5. Local efficience (dismbuned by fla premises costs Totage Postogoging Office Supplies Postogoging Office Supplies Prance services IT support services	rchitecture fo oject is to devo anagement of cont2008 279774 279974 4 cont2008 679 183 18 2997 183 2997 4 cont2008 679 183 183 2997 4 2009 193 2997 4 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 2009 193 193 2009 193 193 193 193 193 193 193 193 193 19	r small and elop suppo nities, in order ? the indivi- project org 33597 33,597,60 8155 220 222 30 1,627 366 432	d medium- ort-systems derta assist- idual crisis). cost 2010 33597.00 33,597.00 815 220 222 30 1,627 366 422	cost 2011 0 0.00 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	Total cost 55191.4 95,192.49 Total cost 2,309 40 55 4,610 1,042 1,224					
Erma (E sized cor The mai tailored t them in manager 9.) work wp4 wp4 wp4 wp4 wp4 wp4 wp4 wp4 wp4 wp4	Electronic Risk Management A munifies). in objective of the ERMA-pro to the needs of small and medium the creation and efficient m ment tasks they are in charge of Project Finance 1. Staffing costs includingsocial county governe TOTAL 6. Local efforces on (dismithened by fin mit) premises costs Talaphone Fostage Postageorging Office Supplies Fostageorging Office Supplies Fostageorging Office Supplies Fostageorging	rchitecture fo oject is to dev n-sized comman anagement of (http://erma.g 27997.4 27997.4 27997.4 27997.4 10 10 10 10 10 10 10 10 10 10 10 10 10	r small and elop suppo nifies, in rof the indivi- project.org 33597 33,697.60 815 220 222 30 915 225 220 222 30 30 31 225 220 222 30 30 30 32 30 30 32 30 30 30 30 30 30 30 30 30 30 30 30 30	d medium- ont-systems der to assist idual crisis). cost 2010 33597 33,597.00 015 223 33,597.00 015 223 30 1,657 355 355 223 30 223 30 1,657 355 223 30 223 30 223 30 223 30 223 30 223 30 223 30 355 355 355 355 355 355 355 355 355	cost 2011 0 cost 2011 0 0 0 0 0 0 0 0 0 0 0	Total cont 951914 9519149 70181cont 2,300 62 83 62 83 4,610 1,042 1,224					

Appendix 3 National Defence Committee Proposition Report; Innst. S. nr. 85 (2008-2009) (Available in Norwegian only)



Innst. S. nr. 85 (2008-2009)

Innstilling til Stortinget fra forsvarskomiteen

St.meld. nr. 22 (2007-2008)

1. SAMMENDERAG J. Inside and the presenter i dense meldingen er foldet i Registingens oppfølging av målaelningen i Socia Merise-tillæringen om et feldsoffig sikkerhei-issten Merise-tillæringen om et feldsoffig sikkerhei-stende i Registingen var en oppfølging av Sikherhei-versent i Registingen var en oppfølging av Sikherhei-ungen for ander sikkerheiter var en effekting sikkerhei-sigen (NOU 2002):44 Farsikter underfahren opp Sor-ingen for ander sikkerheiter og sikherheiters-sigen (NOU 2002):44 Farsikter en enddingen opp Sor-ingen familie ander sikkerheiter. Registeringen gefors i dense meldingen opp Sor-riser binnet. S. n. 9 (2002-2003), jd. Strendel nr. 17 (2013-2002), gefor varksonnetters og justikkerheiter-sikkerheiter og sold berechtag. Bisterikter og sold berechtag. Bisterikter og sold berechtag. Bisterikter og det i sieret og på balanse melden utbere og sold berechtag. Bisterikter og det i sieret og på balanse melden utbere og det i sieret og på balanse melden utbere og det i sieret og på balanse melden utbere og det i sieret og på balanse melden utbere og det i sieret og på balanse melden utbere og det i sieret og på balanse melden utbere og det i sieret og på balanse melden utbere og det i sieret og på balanse melden utbere og det i sieret og det mende av stankfor-sor og det i sieret og det sold mende av stankfor-sor og det i sieret og det genomdelinger og det i sieret i singer melde on en esterni i kreektadetingsreekt i singer melde on en esterni i kreektadetingsree i singer melde on en esterni i kreektadetingsreekt i singer melde on en en en i singer melde on en en i singer i kreektadeting genomdelinger of sold berektag.

 Institling fra forvænksmitter og samenfallensnikker kerktor - samsfrær og samenfallensnikker Til Stortinget
 St. meld. en. 17 (2004-2007) Flodbølgekazumerón i Sar-Asia og semenfallerinshabering, alatet Stortinget utalasket og des peretelle forskababering i døpane-metalens, hennader etableting av en kinesuteetalet. Løsten etableting av en kinesuteetalet. Som 49 (2004-2007) flodbølgekazumerón i som 49 (2004-2007) flodbølgekazumerón som 40 (2004-2007) flodbølgekazumeró

2018 53

55. Regjeringen vil også frømme en ogen melding om brannikkerhet. Omtalen av brann i denne mel-dingen er derfor i hovedsak avgrennet til den opera-tive delen av brannverænd.

1.2 Meldingress insheld og prioriteringer

1.3 Meldingres inshall og prioriteringre I meldingres hæjnel 2 visse og til regerengen forsalt med neddingen og hvitte prioriteringer des legger vist på jeneden 300-3012. Køptel 3 gir en førstelling av tiltak for å styrke arbeidet med aamfannssikkelte på najslendt, regere reførgelingete for førstedpartementes samede ningardig for å tydeliggere og nægrine samede

Innst. S. nr. 85 - 2008-2009

Trondheim og kompetatisetenteret. Statum på Sta-ram Disse med lemmer øreker også å oppetti-holde Vagle leir i Sandren, og fremmer følgende for-slag:

"Stortinget ber Regjeringen opprettholde dagens kompetansesentra på Gisudisen i Trondheim og Su-rurs, og opprettholde Vagle leir i Sandnes."

Komiteena medlem fra Høyre fonstet-ter dörfor at det gjensonføres en sy helselig kone-terson og komadsaulsyse besen på disse og andre relevante fådstere, som næringslivets og ande- og rød-inspestratose helsen for kompennaetvikling, opp-læring og av elser i nørs samhandling og samvirke, og at Regjeringen melder tilbake til Stortinget om dette på egnet vis. Komiteens medlem fra Høyre fontset-

Komiteen merker seg Regjøringem arbeid med å utrofebehovet for tilfuktøren, og avventer de vurderinger og konklusjoner som Regjeringen vil komme med.

komme med. Komit e n er oppatt av at vanling av avidbe-folkningen er et av de sentra (t. tradigotelle avidbe-rvarttiltakene. Sivilforsvarets landidekkende offen-tige varlingssystem besikende av yfstore, samt NRK sin pålolegende til i even formåder av viktige meldin-gør til befolkningen via radio.

bit piedotagenet inter-torinominare av vikuge medninger at herbölangen via rako gör at herbölangen via rako törör att medidangen via sako beröken er som er som er som er som er som er som er komminalisasjösstöknölögi. Flerra her via ander kommalisasjösstöknölögi. Flerra her via kommalisasjösstöknölögi. Flerra her via kommalisasjösstöknölögi. Flerra her via kommalisasjösstöknölögi. Kommalissä kom her som er förstöknör via som er som er som er berekonset er bis Samfannassköknet og Brendsa (DSA, har beslämsgi via via sökningen via som er som er sökningen er som er berekonset er som er som er som er som er berekonset er som e

befolkningen ved bruk av moderne teknologi er et liverddande og skadebegrønernde virkenniddel som ogid Norge må ta bruk. Flert allet ber om at tu-tidegortemendet snæret i spingerter utgrøving og klægjøring av kravpesifikasjoner for vilg av system for moderne befolkningeværling og mölder tilbåle til Stortinget på egnet vis.

37

Komiteens med lemmer fra Arbeidet-partiet og Sosialistisk Venstreparti or kom med ar det i dag finnes teknologiske koninger som kunse vener et gødt rupplænest det skjar en er tilhad med at Regjernigen vil vudrer om tyfor-varlinge abre ensantes med atteen kommenkæjons-teknologi. Disse med langer

teknologi. Disse meditemmer vil påpeke at det nebeides både internasjonalt og nasjonalt itt. befökningsvan-lingabæninger. Disse med den mor vil våder påpeke at departementa atbeider med flere prosjek-te, blatt annet et prosjekt kak Villen Atie-vansmend Fylkesmannen i Sogn og Fjordane. Disse meditemmer vil at atbeidet med beföknings vanling ved slykker, krister og katate-fer forsterner og ber Regjøringen om å komme tilbake til Støringet når det er bessiktamessig.

K o in ite en bar Regjoringen ta med seg erfant-gene En kens ævelse Tyr i arbeidet med vurdering av annen stehnadigt for vanslig av sivitefoldsingen. K om ite en er mig med Regjoringen i at dagena ov om Sivitforwaret ikke gri en korrekt eller tri-attekkelig bjørnmel for Sivilfonsværets virksomhet og organisation.

ber om Stotthererken kens and stotte stotte en stotte stotte stotte en er positiv til at Regjeringen nå er i med a bygget ut an ædnett som blør felles för allt andet attens, og merere dreit vil å syrke andettatenen singlighet för i kommunisere direkte med herrandet, kom i stera deler synet önn at det nyr endformalt med bigget ut at soken er en soken

Appendix 4 National Vulnerability and Emergency Planning Report for 2010 (Available in Norwegian only)



Appendix 5 Samfunnssikkerhet, issue 3 2009 (Available in Norwegian only)



KRONIKK Av Haavard Stemvand, fylkesberedskapssjef i Sogn og Fjordane.



Nye måtar for befolkningsvarsling

Korleis skal publikum varslast om farar og ulykkeshendingar på ein mest mogleg rask og sikker måte? Dette er ein av tinga det skal sjåast nærmare på i det treårige internasjonale prosjektet "Clim-ATIC", som vi frå Fylkesmannen i Sogn og Fjordane deltek i.

"Systemet vil ög kunne nyttast til å formidle informæjon som ikkje er direkte relatert til faresituasjonar"

pel tet er olefinanzier i genom EU st. ta tan bern Perighery Progenamer ("Pylip, og har el deitskarar fri Sverige, Finked, Stottland in revland. I tillegg til Fylkersmannen, er ög and i progistat. fri athæremblet med recosisket er å vise kodels T

cale og regionale styresmak ter kan gjenmfere andynar av klimnsdebartheit og utvikle atøgist for klimntifikesing. Prospisient tekesielt fart i felta berediskap, energi, transport reteielle. Ulike demonstrasjonsprocjekt skal e koteks iskal klimatifikassing kan gå ferer seg reksis.

EFFECTIV DEFOLK MMS GWARSLING Det er og drag symm til å for at firmstådkinnast gl auka fare for uanskte caturhendingar, som denses skred og flaum. Ei effektiv befokknings vansling vil difor vere eit viktig verknay for å bladene at är, belere og materielle verding går ta når släke situasjonar oppstår. Dette set store k

> arslinga er å sikre at den år fram til flest mogleg, g syte for at også dei ler mest sårbare vert Md^a

til værslingssystemet, då ein må vere trygg på værelet innæn svært kort tid når fram til alle r oppheld seg i det akroelle omsådet. Som ein del av demonst næjonsprosjektet er de

tenkt gjernomfært el follskala vamlingsæris byrjinga av 2010. Det kan være aktuelt å var alle som bur eller oppheld seg i ein av komm nane i fylket. Værselet skal sendast som tek talemsédingur til böde fara- og mobilheisforur, ved hjelp av teknologi for lokasjondusert befolkaingsvarstilge. Fölkt varslingssystem vi skreat varselet ikkje berer hagar opp personer med registrert adresse i omsådet, som at det ög når fram til andre som er der met tilfeldig.

Det hat i dei seinare drivore gienzomfart, og er under planlegging, fleine avinger der telefonbuetet befolkningsvanling er eit av eringemomerta. Under desse avingane bar vanlings likevel berev vone ein avgressen det av den totale avinga. I vist grosjekt er be höltningsvanlinger

vil ditor vere svært viktig å gjennomfære el god evakering. Evakeringsmit kanne sele om varselet faktisknåtde fram til, og vært forstätt av miljenspese. Evakeringsærbeldet kan viktare vere med på å Genettissere værsmåte sterinisk forbeiringspor enskel og viktige andre uppfølgingspunkt. Hag

 Bit poeng for projektet er å sjå om det kan ebberat vandingssystem med el "autilitekte"tikowning, ikk at føre slåk estatur/nyssmakter samabelder om etableringa av ett føler system til å snede værst om difa typer rikit og

Systemer via big kanne system til å formålle informanjon som klikke er inskne rekonstrut filterultraniganas. Det kan til dennes svere informasjon til trafikatarase ver klostoneksystige i langeløkt, eller tranging av kommunal ditkke-ranforsyslagan nå der trafikatarase verklande som klikkeranforsystemer for elle breite brukksommåde, vid operatorene er systemet få verdfull mering i må, og effektiv tradensiga versel. SOGALE MEDIER Er vikig moment wed vandings er å sikre ei di aller mest skråve ver indså. Svært menge kår aller mest skråve vert nådå. Svært menge kår henkar enno likkje mobiliteletor, og sveksyrstna poblem med aller visitsredeliger. I øving vil det difter vere vikigt ag tip å behvært forst bernade verlike genom sindve verligetar

"Ei effektiv befolkningsvarsling vil difor vere eit viktig verktøy for å hindre at liv, helse og materielle verdiar går tap"

> et vil og vere interessant å sjå på om ostale medler", som til dentes Facebook, blogger Twitter, kan systaet som kanalar for formkling varset og sappletande informasjon.

noment i prosjektet or å vandere kva etat so I gjete jobben med å sende ut varselet. Både ritsentralen (110), politier (112) og behefare

taket (113) er med i referansistrup det tet. Det viel vere namring at ein av i det far oppgles med den prak tiske var av sig sjøv eller ander. I referanse en oppsessentarat for er inskåg andre, opp private detears som til dennes DSR, vepvisser. Verstamhforsking Hag ander, Fjohane og Antifand komstamler.

> rands arbeidet mol vurdering av betekningsäng Prosjekter vart omtak i innettillinga frå varkoniteret i suka "Samtlumentikkerhetvicke og samtehning". Komisidierinalet strabe om anna at"...effektiv varhlige av betekkninved brak av moderne takninkog er et liveredde og skadebegensende vitkemiddel som også gen at så brak...".

Sadars prozjekani inngår i elt startre, imærnasjonalt prosjekt, vil det ög vere viktig ä bådra til internasjonal enfaningvet veksling og spreising av kunnskapar ved prosjektskatt.

Roms Sogs og FAKTA Ist det informasjondstræunen me ktillige myndigheiter, medium og bi

Informasjonistriaumen mellom myndigheiter, medium og befolkninga al sentruk betyding i krisenituasjoner. DSB har fått i oppdrag av Justis-

Addressing by the under og etter knoen let innberer både evundering av ikke atternativ for ett landsdekkande system for børdkningevanling, samt system og organisering som kall skire ensklapige kommunk agen melon myndigheter og opterfolkning. Det sverordna målet er å kunne handt ere kohrandname skulandet die

ordna målat er å kunne handtere raordinære attuegjoner på en e om reduserer tap av lv, helse, og materielle verdiar

Appendix 6 Statistics from the online evaluation form (Available in Norwegian only)







Copyright © 2002-2007 more as - Clim-atic - Varsingseving 2010 - side 3







synght & 2002-2007 more as - Clim-ATIC - Exstremwire-twarslingsaving - side 2

2002-2007 more as - Clim-ATIC - Ekstremvér-lva side 1









Ja Er du i Aurland? Alternativ Ja Nei Tal svar 24 0 Svar-del 100,0 % 0,0 %

Copyright © 2002-2007 more as - Clim-ATIC - Ekstremvér-Ivanslingsaving - side 3

Copyright © 2002-2007 more as - Clim-ATIC - Ekstremvér-Ivansingsaving - side 4



Copyright © 2002-2007 more as - Clim-ATIC - Ekstremvér-Ivarslingseving - side 5

Copyright © 2002-2007 more as - Clim-ATIC - Ekstremvér-Ivarsingsoving - side 6